RESEARCH Open Access



# Attitude and intention of migrant populations in the Netherlands regarding female genital mutilation/cutting

Ramin Kawous<sup>1,2\*</sup>, Emily Allwood<sup>2</sup>, Annemarie Middelburg<sup>3</sup>, Anke van der Kwaak<sup>4</sup>, Diana Geraci<sup>2</sup>, Marthine Bos<sup>5</sup> and Maria E. T. C. van den Muijsenbergh<sup>2,6</sup>

# **Abstract**

**Background:** Due to migration, the practice female genital mutilation/cutting (FGM/C) has become an issue of concern in high-resource countries such as the Netherlands. It was therefore of utmost importance to explore the attitude and intention of migrant populations in the Netherlands towards FGM/C, which may be leveraged to promote its elimination. Therefore, the aim of the present study was to explore the attitude and intention of migrant populations in the Netherlands regarding FGM/C.

**Methods:** A qualitative study design was employed using Theory of Planned Behaviour (TPB) as a framework for the analysis. Data were collected using focus-group discussions (FGDs) and individual interviews. The FGDs and individual interviews were audio-recorded and transcribed *verbatim*. The main topics of the interviews were based on the constructs of TPB (attitude, subjective norms, perceived behavioral control and intention). Thus, concerning the development of categories, we opted for a hybrid form using a deductive as well as an inductive approach.

**Results:** A total of 55 participants, 15 men and 40 women (9 born in the Netherlands) participated in the study. The findings showed that as a result of migration and regardless of country of origin and gender, many participants have changed their attitudes towards the abandonment of FGM/C. None of the participants intended to have FGM/C performed on their daughters. Generally, the social pressure to perform FGM/C seems to be lower in the Netherlands when compared to the country of origin. Most participants felt confident in their ability to resist social pressure. However, some participants feared that they might succumb to social pressure or feared that their daughters would undergo FGM/C without their consent.

**Conclusion:** This study aimed to explore the attitude and intention of migration populations in the Netherlands regarding FGM/C. Our findings showed that the study participants had no intention to perform FGM/C on their daughters. As a consequence of acculturation process, interest in the practice of FGM/C could wane following migration. Nonetheless, some pressure to perform FGM/C still exists after migration. Newly arrived migrants and those more vulnerable to social pressure, may benefit from educational interventions that increases knowledge and awareness about various aspects of the practice, with an emphasis on empowering those individuals in facing social pressure.

**Keywords:** Attitude, Female genital mutilation/cutting, Intention, Migrants, Social pressure

<sup>&</sup>lt;sup>1</sup> Erasmus University Medical Centre, Rotterdam, The Netherlands Full list of author information is available at the end of the article



© The Author(s) 2022. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third partial in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

<sup>\*</sup>Correspondence: r.kawous@outlook.com

Kawous et al. BMC Women's Health (2022) 22:405 Page 2 of 11

# **Plain English summary**

Female genital mutilation/cutting (FGM/C), also known as female circumcision, is a practice which involves the cutting away of part or the whole of the external female genital organs. The practice has no health benefits, and it harms girls and women in many ways. The migration of girls and women to high-resource countries such as the Netherlands, has led to the development of programs to prevent FGM/C and to care for those affected by FGM/C.

The purpose of this study was to explore the attitude and intention of migrant populations in the Netherlands towards FGM/C. The results of this study may be relevant in the development of interventional programs to promote the elimination of FGM/C. Our findings showed that the study participants had no intention to perform FGM/C on their daughters. Many participants may have changed their attitudes towards the abandonment of FGM/C. The social pressure to perform FGM/C seems to be lower in the Netherlands when compared to the country of origin, however, some pressure to perform FGM/C still exist after migration. In this regard, educational programs are necessary, with an emphasis on empowering vulnerable groups such as newly arrived migrants in facing social pressure.

# **Background**

Female genital mutilation or cutting (FGM/C) refers to a cultural practice which involves 'partial or total removal of the external female genitalia, or other injury to the female genital organs for non-medical reasons' [1]. The World Health Organization has classified FGM/C into four major types [1] (see Table 1).

FGM/C is traditionally practiced in several countries in Africa, Asia, and the Middle East. As a result of international migration, the practice of FGM/C has become an issue of concern in high-resource countries, such as the Netherlands [2]. The reasons for practicing FGM/C, the extent of the practice, and type of cutting vary across ethnic groups and by geographic region. It is widely documented that justifications for supporting the practice of FGM/C are numerous and in their specific context, compelling. The practice is rooted in gender inequality and attempts to control women's sexuality and to protect the honor of the family by ensuring virginity until marriage and martial fidelity. In addition, FGM/C would confer girls the status of eligibility for marriage [1, 3]. Globally, it

**Table 1** World Health Organization classification of female genital mutilation/cutting [1]

Туре	Definition					
I	Partial or total removal of the clitoral glans and/or the prepuce/clitoral hood					
II	Partial or total removal of the clitoral glans and the labia minora, with or without removal of the labia majora					
III	Narrowing of the vaginal opening through the creation of a covering seal by cutting and repositioning the labia minora and/or the labia Majora, with or without removal of the clitoral prepuce/clitoral hood and glans					
IV	All other harmful procedures to the female genitalia for non-medical purposes, e.g., pricking, piercing, incising, scraping and cauterizing the genital area					

is estimated that more than 200 million girls and women have undergone FGM/C [4] and nearly 68 million girls are estimated at risk between 2015 and 2030 [5].

The practice has been associated with negative health outcomes, and those with FGM/C can suffer significant, long term, and irreversible physical, psychological, and psychosexual complications [6–11]. Because of the harmful consequences to women's health and wellbeing, and the fact that FGM/C reflects deep-rooted inequality between the sexes and constitutes an extreme form of discrimination against women, the practice is internationally recognized as a human rights violation [1]. Globally, many countries have increasingly undertaken law reform to prohibit the practice [12, 13].

It is estimated that about 41,000 girls and women with FGM/C are living in the Netherlands and over 4,200 girls are at risk of being cut in the next 20 years [14]. Performing any form of FGM/C is forbidden by law in the Netherlands and punishable by a prison sentence. The Dutch approach against the practice of FGM/C focuses on preventing the practice through engagement of communities and training of (healthcare) professionals [15]. In the Netherlands, regardless of ethnicity, all children under 19 years receive general medical examinations - including genitalia up to age four—as part of a standard preventive healthcare approach performed by Youth Health Care providers. These professionals are also required to identify and assess the risk of FGM/C among migrant populations. These professionals inform parents of the health consequences of FGM/C and explain that the practice is prohibited by law. Additionally, parents are provided with a document about this law ('statement opposing female circumcision'), signed by official authorities, to support parents in resisting pressure to have FGM/C performed when traveling abroad [16].

As a result of increasing migration, FGM/C has become a growing concern among policy makers and (health)professionals in high-resource countries such as

Kawous et al. BMC Women's Health (2022) 22:405 Page 3 of 11

the Netherlands [14, 17, 18]. While the attitudes of practicing populations may be leveraged to promote its elimination by developing effective policies and intervention programs, there remains a lack of knowledge regarding the attitude and intention of migrant populations in the Netherlands towards FGM/C. Moreover, professionals need to know about the practice of FGM/C in general, and understand the social dynamics that perpetuate FGM/C after migration, in order to be able to identify girls at risk and to provide culturally sensitive care to women with FGM/C.

Given the traditional nature of FGM/C, the practice has strong positive social and cultural value, which makes it difficult to eliminate the practice. Understanding factors associated with performing FGM/C could be helpful to develop effective interventions to change this behaviour. Therefore, the aim of the present study was to explore attitude and intention of migrant populations from FGM/C practicing countries in the Netherlands regarding FGM/C.

The study design was inspired by the framework of Ajzen's Theory of Planned Behavior (TPB) [19]. The TPB is one of the most prominent theories in behavior change.

The TPB proposes that behavior is predicted mostly by intentions to engage in behavior, and intentions are in turn predicted by three main variables: attitudes toward the behavior, perceived behavioral control, and subjective norms. Attitudes refer to individuals' positive or negative evaluation of the behavior; perceived behavioral control refers to perceptions of whether the behavior is within our control; and subjective norms (social pressure) refer to perceptions of whether the 'entourage' (friends and family), important others, or the reference group approve of and themselves engage in the behavior.

In this study, we were particularly interested in exploring factors that may contribute to migrants' intentions towards FGM/C. It is important to note that in this study, our aim was to explore the constructs of the TPB, rather than to examine the predictive value of these constructs in relation to migrants' intentions to have FGM/C performed on their daughters.

Previously, most theoretical frameworks have been used to understand FGM/C at the community level (e.g., society and communities), such as Social Convention Theory [20], the Community Readiness Model [21], and the Diffusion of Innovation Model [22]. However, similar to Mackie and LeJeune [20] and Barrett and colleagues [23], we are of the opinion that beliefs and norms are better understood not only at the community levels but also at the individual level. The TPB combines elements from theories of individual behavioral change and of theories that regard FGM/C as a social convention [24].

## **Methods**

### Study design

A qualitative study using seven semi-structured focus group discussions (FGDs) with migrant populations from FGM/C practicing countries was carried out between June and October 2018 in the Netherlands. The Interpretative phenomenological analysis (IPA) was used to capture the experiences of the participants. The IPA focus explicitly on the links between how people describe their experiences, their cognition and their behavior. The IPA was deemed the most appropriate theoretical approach to data collection and analysis in order to understand participants' experiences and perceptions regarding FGM/C.

Focus groups were considered an appropriate method of data collection for in-depth exploration of perspectives of participants on FGM/C, as it encouraged them to share their perspectives openly and stimulated interactions between them. Focus groups have been previously successfully used to collect data on FGM/C [25-27] and other sensitive (health) issues (e.g., [28]). In addition to the FGDs, five individual interviews were conducted to allow in-depth exploration and to ensure thematic saturation. These interviews confirmed that data saturation had been reached. This paper reports the results of this study in line with the Consolidated Criteria for Reporting Qualitative Research (COREQ) criteria. The COREQ is a checklist developed for promoting explicit and comprehensive reporting of qualitative studies and to 'indirectly' improve rigor, comprehensiveness and credibility of such research [29].

The study protocol was approved by the Medical Ethical Committee of Erasmus Medical Centre, Rotterdam, the Netherlands (MEC-2018-2216).

# Sampling and recruitment

Using purposive sampling, fifty-five male and female participants both first and second generation from countries where FGM/C practiced were recruited through trusted intermediaries. Of these, two participants who participated in one of our previous studies on migration and health were reapproached for participation. There was no relationship between the research team and the participants prior to commencing the study.

The recruiters were instructed to recruit participants with as even a distribution as possible among age range and gender and those who have not participated in a study on FGM/C. Unfortunately, due to limited time and resource, less male participates were recruited than female participants. The Netherlands has a significant number of migrants from very high FGM/C prevalence countries such as Somalia and Eritrea. Therefore, more

Kawous et al. BMC Women's Health (2022) 22:405 Page 4 of 11

people were recruited from one of these countries. The recruitment was stopped when data saturation was reached.

### Study participants

The 55 participants in this study included 15 men and 40 women from countries where FGM/C is practiced, including 9 young women born in the Netherlands to parents from an FGM/C-practicing country (so called 'second generation', in comparison to the 'first generation' participants born in their country of origin'). Ages of the participants ranged from 18 to 70, with a median age of 42 years (see Table 2). The discussions lasted between 74–120 min.

In order for the participants to feel comfortable, we deliberately did not ask about their FGM/C-status; however, during the FGDs and interviews some women mentioned being cut. The study participants seemed to feel at ease talking about their experiences.

The average number of children in each family was 3, with one woman having 8 children. The median age of their children was 15 years. The average length of stay in the Netherlands was 12 years. The majority spoke Dutch well and were either in education or employed. Most participants frequently kept in touch with their families in their country of origin, mostly by phone or social media,

and they had visited their country of origin at least once since arrival to the Netherlands. Some participants were accompanied by their children during their visit to their country of origin.

### Data collection and setting

The interview guide was inspired by the framework of the Theory of Planned Behavior (TPB) [19]. The (interview)topics included: (1) participants' attitudes towards FGM/C, (2) perceived social pressure from others to perform FGM/C (subjective norms), (3) participants' confidence in their ability to prevent their daughters from undergoing FGM/C (perceived behavioral control), and (4) their intention to have their daughters undergo the practice. The (almost)similar topic list for both the FGDs and interviews, was continuously fine-tuned and tailored throughout the data collection process, according to insights gained during the FGDs and interviews.

The FGDs and individual interviews were conducted in Dutch, with the exception of one FGD- and two individual interviews with participants from Eritrea which were conducted in Tigrinya with the help of professional interpreters, as the researchers did not speak Tigrinya.

As socio-cultural factors may inhibit speaking out in the presence of persons of the other sex, the FGDs

**Table 2** Characteristics of the participants

	Focus group disc			Individual interviews		Total	
	Women 1st and 2nd generation	Men	Women, 1st generation	Women, 2nd generation	Women, 1st generation	Men	
Number of FGDs Date and duration (in minutes) of session	[1] Jun., 21 99 min	[2] Jul., 14 74–93 min	[3] Jul., 7 and 21; Sep., 21 109 -119 min	[1] Oct. 20 117 min	Sep., 8 66–75 min	Sept., 8 68 min	[7] June–October 2018
Number of participants	3	14	25	8	4	1	55
Age range	23-50	32-70	19–56	18-29	45-50	54	19–70
Number of women mention- ing being cut	2	N/A	7	0	2	N/A	11
Do you have children?							
Born in the country of origin	3	15	76	0	16	0	110
Born in the Netherlands	6	13	17	0	10	0	46
Country of origin (Number of participants)	Somalia (2), Guinea (1)	Somalia (3), Ethiopia (2), Eritrea (2), Sudan (3), Egypt (1), Togo (2), Ghana (1)	Somalia (6), Egypt (1), Sudan (2), Sierra Leone (1), Eritrea (15)	Somalia (5), Eritrea (2), Egypt (1)	Somalia (1), Eritrea (1), Iraq (1), Egypt (1)	Eritrea (1)	Somalia (17), Guinea (1), Ethiopia (2), Eritrea (21), Sudan (5), Egypt (4), Togo (2), Ghana (1), Sierra Leone (1), Iraq (1)

Kawous et al. BMC Women's Health (2022) 22:405 Page 5 of 11

were conducted among men and women separately, using same-sex moderators. During the interviews, the term 'female circumcision' was often used because it is the most acceptable term within the participants' communities.

FGDs were led by four moderators (RK, EA, AvdK, MB). The moderators were either researchers or service providers with experience in FGM/C. They were accompanied by an assistant who was also an expert in the field of FGM/C. The assistants were responsible for taking notes and audio recording. The individual interviews were conducted by RK, MB, and DG.

The interviews and FGDs were held in settings that ensured privacy, most often at conference centers near the main train stations in the Netherlands. Participants were provided with lunch and refreshments before the FGDs to make them feel at ease and to create a safe space. Researchers also offered travel reimbursement and a  $\ensuremath{\epsilon}$ 35 gift card to acknowledge their contributions.

All interviews and FGDs were audio-recorded and transcribed *verbatim* in Dutch.

# Data analysis

All interviews were recorded, transcribed, anonymized and uploaded into the computer software package ATLAS.ti (http://www.atlasti.com, version 8). Data were analyzed in two stages using both (I) deductive and (II) inductive analysis methods. The TPB served [19] as the explanatory background guiding the interpretation, understanding and coding of the data material. Therefore, in the first stage of data analysis, a deductive approach was applied whereby key categories were pre-determined according to the elements of the TPB (attitude, subjective norm, perceived behavioral control and intention). In the second stage, the transcripts were read through again to search for more features and patterns. An inductive analysis was then performed to identify new themes that may be not captured in the framework used.

# Trustworthiness and rigor

To ensure the trustworthiness of the data, approximately 30% of the transcriptions coded by RK were also coded by EA. Multiple coding can be a valuable strategy to assess inter-rater reliability and enhance interpretations or coding frameworks [30]. Initial differences were discussed until agreement was reached. Participants were also asked for any additional comments and/or clarifications if something was unclear. In addition, immediately after the FGDs and interviews, the moderator and assistant moderator debriefed and compared notes.

### **Results**

### **Attitudes**

# Participants' attitude towards the (dis)continuation of female genital mutilation/cutting

The participants were asked about their views on FGM/C and to describe the importance of FGM/C. Regardless of country of origin and gender, most participants do not support the continuation of FGM/C. Different reasons were given by the participants for not supporting the practice. They mentioned, for example that FGM/C causes severe health problems, the practice has no benefits, FGM/C is not in the best interest of the child and that it is not a religious requirement. Particularly, young female participants born in the Netherlands, described FGM/C as child abuse, and further contended that the practice was a violation of children's rights since it is performed without their consent. For example, a 24-year-old female FGD participant born in the Netherlands (to Egyptian parents) said:

"I feel that it is an injustice to subject girls to FGM/C."

Study participants noted that the significance of FGM/C has been changing, both in the Netherlands and in their country of origin, from being seen as a necessity to a practice that is being questioned and abandoned. Both men and women talked about the practice in terms of how FGM/C 'used to be' or how FGM/C was 'in the country of origin'.

"Upon arrival in the Netherlands, it felt obvious that I would circumcise my daughter in the future, but when I learned more about the practice and about the consequences attached to it, I knew for sure that I wouldn't subject my daughter to FGM/C, it just didn't feel normal..."

A 19-year-old female FGD participant from Somalia. In one of the FGDs, female participants from Eritrea mentioned that they had experienced campaigns against FGM/C in the country of origin focusing on the harm of FGM/C and that these campaigns encourage people to question the practice.

In general, due to the sensitive and taboo nature of the practice, FGM/C is not talked about between individuals, including partners (in marriage) and their children. However, some participants noticed a change in behavior and in the discourse on FGM/C: from barely being topic of a conversation in their country of origin, they had started discussing the subject after their migration to the Netherlands. The female participants born in the Netherlands mentioned that FGM/C can be discussed between young people, but they found it difficult to discuss it with older generations.

Kawous et al. BMC Women's Health (2022) 22:405 Page 6 of 11

"I have met recent migrants [in the Netherlands], I felt that it was very difficult to talk about female circumcision with them. They had young girls, circumcised, and they considered it part of culture." A 23-year-old female FGD-participant born in the Netherlands (to Somali parents).

In general, there was a lack of knowledge and awareness among female participants born in the Netherlands regarding FGM/C. Some of them indicated that they did not know much about FGM/C before they participated in this study—they either searched online or asked their mothers about the practice.

Concerns were shared across participants about newly arrived migrants from countries where FGM/C is practiced having positive attitudes regarding FGM/C. They added that new groups need time to change their views towards the discontinuation of the practice.

# Role of adverse health consequences in the discontinuation of FGM/C

The health risks associated with FGM/C were mentioned as an important reason for supporting the discontinuation of the practice both in the country of origin as well as after the migration. According to most participants, regardless of country of origin and gender, the consequences for a girl or woman undergoing FGM/C were understood as predominantly negative. The extent of knowledge about the health consequences of FGM/C varied among participants—some had either experienced, witnessed, or were familiar with the related health consequences.

"I became familiar with the health consequences of FGM/C after migration [to the Netherlands], and I think differently about it [in a negative sense], I didn't know anything about it back home [about the health consequences of FGM/C]"

A 50-year-old female FGD participant from Somalia. Participants mentioned several psychological, sexual, and physical complications related to FGM/C. The impact of FGM/C on a woman's capacity to enjoy sexual intercourse, pregnancy, and childbirth was a substantial issue for some participants.

"...it [FGM/C] had a long-term effect on my psychological wellbeing, it was painful being forced to undergo the procedure... I had psychological problems and I went to a psychologist a couple of times. I dreamt about it [FGM/C], probably because I thought about it a lot. I find it terrible for girls to undergo circumcision...and I became aware of its complications during my second delivery. A caesarean section was performed with the delivery of my daughter... I stayed 5 days at the hospital, but she was not coming out. This happened again during the delivery of my son."

A 50-year-old female Interview participant from Iraq. The fact that she had been subjected to FGM/C at a younger age was quite troubling for this participant from Iraq.

### Reevaluating the role of religion

Religion was mentioned as an important reason for practicing FGM/C in the country of origin, particularly in Somalia where the religion of Islam is almost universal. Female participants from Eritrea expressed that FGM/C is nowadays mainly practiced in Muslim communities in Eritrea, whereas in the past the practice of FGM/C was also dominant among Christian communities. Most participants from Somalia expressed that it is incorrect to assert that FGM/C is 'Sunna' in Islam; they stated that only male circumcision is 'Sunna' in Islam.

"I have three boys who I had circumcised, but who was I to decide to circumcise my boys, I could have left the choice up to them until they were capable of deciding for themselves...If I think about it now, if God intended men to be cut, he would have made it like that."

A 55-year-old female FGD participant from Eritrea. Across the FGDs, male circumcision was often brought up when participants were asked about FGM/C. They did so to make the comparison between the two practices. This female FGD participant from Eritrea (55-year-old) who migrated in the early nineties to the Netherlands, even questioned male circumcision, and pointed out the

# **Subjective norms**

right to bodily autonomy.

# The role of social pressure on the continuation of female genital mutilation/cutting

In the context of this study, subjective norms refer to perceived social pressure from people in the Netherlands or in the country of origin to perform FGM/C. Across the FGDs and interviews, it was underlined that the social pressure and cultural significance of FGM/C in the country of origin is different compared to the Netherlands: social pressure for subjecting girls to FGM/C seems to be overall lower in the Netherlands when compared to the country of origin. After their migration to the Netherlands, they did not risk the social acceptance of their daughters nor were they concerned about the marriageability of their daughters. Moreover, with legislation outlawing the practice in the Netherlands, they felt more supported to resist social pressure to perform

Kawous et al. BMC Women's Health (2022) 22:405 Page 7 of 11

it. Nonetheless, the pressure to continue the practice of FGM/C emerged in some of the FGDs. A 42-year-old male FGD participant from Sudan said:

"One of my relatives, she is actually my aunt, talked with me about this [FGM/C] [by phone]. She said that they were circumcising their youngest son next week and asked me 'what about your daughter, [name]? 'I replied, 'What about her?' I said, 'No, I will not do this to her, it is a very bad thing."

Some participants were afraid that their daughters would undergo FGM/C without their consent, for example during a holiday in the country of origin. Several participants, mainly from Somalia, Eritrea and Sudan, expressed that at some point after migration to the Netherlands they had experienced pressure from mothers, grandmothers, aunts, or mothers-in-law to have FGM/C performed on their daughters.

# Marriageability, relationships, and female genital mutilation/cutting

In the countries where FGM/C is practiced, marriageability is mentioned as one of the important reasons for supporting the continuation of the practice. Our participants described how cultural and societal norms place pressure on their communities in country of origin to perform FGM/C, to prepare their daughters for adulthood and marriage. According to participants, the practice is often considered a traditional rite of passage to womanhood, decreasing girls' sexual desire, and thus protecting virginity, which is seen as prerequisite for marriage in the country of origin. However, while on average uncut girls may be considered ineligible for marriage in countries where FGM/C is practiced, our several female participants mentioned that men from FGM/C practicing-countries in the Netherlands prefer uncut girls (from own origin group) as their brides, especially girls who are not infibulated.

"...in my social network, most Somali boys say that they do not want to marry a circumcised girl, because of its negative [health] consequences...when I wanted to marry my husband [fiancée in the Netherlands], he said that he was not planning to marry me if I was infibulated...it felt intense [painful], everything felt gone, love, everything..."

A 39-year-old female FGD participant from Somalia. Some female study participants also mentioned that although people often prefer to marry those with a similar background, they noticed that people are starting to establish relationships with those from other countries than the country of origin.

# Decisions and intentions to perform female genital mutilation/cutting

The participants were asked about cases of FGM/C being performed in the Netherlands and if the people living in the Netherlands intended to have their daughters cut if there was, for example no law that criminalized the practice of FGM/C. Regardless of country of origin and gender, none of the participants had the intention to have FGM/C performed on their daughters and participants were not aware of actual FGM/C cases performed in the Netherlands. However, anecdotal examples about cases of FGM/C performed elsewhere in Europe, or girls that were taken to their country of origin to undergo the practice, were provided in some FGDs; there were rumors about "cutters" that were flown into the Netherlands. Across the FGDs, some participants mentioned that if parents really wanted to practice FGM/C, they would travel during the holidays to their country of origin due to fear of prosecution in the Netherlands and possibly perform less severe forms of cutting, believing that the less severe forms will be undetectable by medical- or legal authorities in the Netherlands. Roughly half of the participants said that they would expect that more people would practice FGM/C in the Netherlands if there was no legislation prohibiting the practice.

Interviewer: "If we take a look at legislation, if circumcision would not be forbidden by law, would we see more circumcision here in the Netherlands? Participant (50-year-old female from Somalia): Of course

Participant (23-year-old female born in the Netherlands (to Somali parents): I think so too"

At the same time, some other participants said that they expected that women who had experienced FGM/C themselves, and who truly understood the consequences of the practice, would know that the practice is wrong and would not let their daughters undergo FGM/C, whether it was prohibited by law or not.

Across the FGDs, some participants mentioned, as an unintended side-effect of the law, that some parents leave their daughters behind in the country of origin after they have been cut, while they themselves travel back to the Netherlands. Overall, the participants believed that legislation is important as it has a deterrent effect, but the mere existence of legislation is not enough to stop the practice.

# Perceived behavioral control

Participants were asked whether they could prevent their daughters from being cut (perceived behavioral control). Several participants from Somalia and Eritrea were very Kawous et al. BMC Women's Health (2022) 22:405 Page 8 of 11

determined in refusing to perform FGM/C and indicated that they would easily resist social pressure from people by explaining their decisions. Social pressure made some participants recall their own experiences, and made them think about this as unjust, which triggered a change in views about the practice, creating a sense of apprehension accompanied by a sense of obligation to protect their daughters.

"When my daughter was born, my mother called me and told me to circumcise my daughter. I thought, maybe she is right. After a while, I completely changed my mind and said 'How could you [think such a thing]?' I said, '...my daughter is born here [in the Netherlands], I don't want to hurt her, because I have been through this myself and I still have nightmares about cutting." A 40-year-old female FGD participant from Somalia.

Some participants expressed fear about their daughters undergoing FGM/C without their consent during a visit to their country of origin. In fact, some said that they would take measures to prevent FGM/C, for instance, their children would not accompany them when visiting their country of origin, and if they were accompanied by children, they would lie about the FGM/C status of their daughters.

"If we are on vacation in our country of origin, I will lock the doors [to protect her daughters] ... If people go there, just to visit [the country of origin], and when you say that you would not cut your daughter, they will then question your way of thinking. They may lie to you and cut your daughter without your consent."

A 39-year-old female FGD participant from Somalia. Across the FGDs, several participants pointed at the 'statement opposing female circumcision—document' (signed by official authorities in the Netherlands) and said that this helped them explain to family in the country of origin that FGM/C is prohibited in the Netherlands and enabled them to resist the social pressure.

### Discussion

This study aimed to explore the attitude and intention of migrant populations in the Netherlands regarding female genital mutilation/cutting (FGM/C). We explored their perceived social pressure to perform FGM/C (subjective norms) and perceived confidence in being able to prevent FGM/C (perceived behavioral control). These factors may be closely interrelated and dependent on each other and are likely to affect individuals' decisions to engage in the practice of FGM/C [31, 32].

Our findings showed that as a result of migration, regardless of country of origin and gender, many participants have changed their attitudes towards the abandonment of FGM/C in the Netherlands. None of the participants intended to have FGM/C performed on their daughters. The adverse health consequences of FGM/C, legislation outlawing FGM/C, that the practice is not necessarily required by religion, and that the practice is as a violation of children's rights were among the most cited reasons by our participants to not support the practice of FGM/C. The social pressure to perform FGM/C is experienced as to be overall lower in the Netherlands when compared to their country of origin. Most participants felt confident in their ability to resist social pressure; however, at the same time some participants feared that they might succumb to social pressure or feared that their daughters would undergo FGM/C without their consent, for example during a holiday in the country of

The study findings echo the results of several other studies, which demonstrated that interest in the practice of FGM/C wanes following migration and that support for the continuation of FGM/C for girls born in the host country is low [33–41]. In a study, Morison and colleagues [33] investigated experiences and attitudes towards FGM/C among migrants from Somalia in the UK. They found that older generations, migrants with few signs of assimilation, and new arrivals are more likely to hold traditional views towards FGM/C. Intentions to have their daughters undergo FGM/C varied significantly with the time spent in the UK—the more integrated people were, the less likely they were to uphold traditional views towards FGM/C.

Support for the abandonment of the practice in the diaspora can be a result of acculturation. Acculturation is a process by which cultural and psychological changes occur through the contact between two or more cultural groups and their individual members [42]. Although migrants carry their cultural heritage with them when they migrate, at some point during their residence the process of acculturation occurs, and they adapt to their host country's culture. It is important to note that acculturation effects may need time to consolidate. There is evidence to show that the amount of time spent in the host country is associated with increasingly negative attitudes towards FGM/C, and that newly arrived migrants may have more positive attitudes towards the practice [26, 37, 38, 43]. This is not surprising since the more newly arrived migrants come from an environment where FGM/C still is a social norm, and not conforming to the norm could result in social exclusion, stigma, and the inability to find a suitable marriage partner [44].

Kawous et al. BMC Women's Health (2022) 22:405 Page 9 of 11

In Netherlands, as in many other host societies, the practice is uncommon and criminalized by law. According to our participants legislation against FGM/C is critical as it decreases support for the continuation of FGM/C among migrant populations. Indeed, to date, no one suspected of FGM/C has been convicted by a court in the Netherlands [45]. The enactment and enforcement of a law criminalizing FGM/C can challenge existing social norms by providing legitimacy to new sets of behaviors [44]. There is evidence to suggest that people may abandon FGM/C in an environment with legislation outlawing the practice, even if they favor the continuation of FGM/C [33-38]. Conversely, we believe that individuals from migrant populations with a more favorable attitude towards FGM/C, those with less or no confidence in their ability to prevent their daughters to undergo FGM/C, and/or those who perceive more social pressure from others, may be more likely to perform FGM/C on their daughters, regardless of the legislation outlawing the practice. As an unintended side-effect of the law in the host country, according to our participants, parents with more traditional views towards FGM/C could even leave their daughters behind in the country of origin, suggesting the fear of not conforming to the social norm may be stronger for some individuals than the fear of prosecution in the host country. It is thus important that legislation is accompanied by other preventive measures to protect girls at risk.

Further, marriageability, indirectly linked to preserving virginity, is to date among the more common or even the primary reason in some communities to perform FGM/C [44]. However, our study participants were not concerned over marriage prospects as a justification for FGM/C. They noticed that men from FGM/C practicing countries in the Netherlands prefer uncut girls (e.g., from own origin group), and that people are starting to establish relationships with those from other countries. This finding could be regarded as a sign of acculturation and in line with previous studies suggesting that migrants from FGM/C practicing countries are often relatively positive about uncut girls [26, 37, 38].

In general, as part of acculturation process, more interactions with members of other social groups in the host country who do not share expectations regarding FGM/C, may contribute to a broader shift in perceptions of societal norms among migrant populations from FGMC practicing countries. For instance, male circumcision is common in several countries where FGM/C is concentrated (40,41); our participants often talked about male circumcision when asked about FGM/C and one of our participants even questioned the practice of male circumcision and pointed out a child's right to bodily autonomy. While this one single case is not sufficient to

make a generalized conclusion, it is reasonable to assume that due to acculturation, shifting opinions among some migrants could go beyond the practice of FGM/C.

Social norms or subjective norms surrounding FGM/C and perceived behavioral control are important determinants of decisions to engage in the practice of FGM/C. While social pressure to perform FGM/C still occurs after migration, similar to previous studies [34, 40, 46, 47], we found that the pressure seems to be overall lower in the host countries such as the Netherlands when compared to the country of origin, and that most migrants report being confident in resisting social pressure. Migrant populations might differ from their counterparts in the country of origin in terms of social pressure. Their counterparts might succumb to social pressure and conform to what they believe others expect of them, however, migrant populations experience overall less social pressure in the host country, hence they may experience less fear of social exclusion for not performing FGM/C. Perceived behavioral control is strongly related to social norms or subjective norms. The more support and the lower pressure to perform FGM/C, the greater individuals' perceived behavioral control and the stronger their intention to not engage in the practice of FGM/C. A large share of our participants were determined to refuse to have FGM/C performed on their daughters and they were not afraid to protect their daughters. The 'statement opposing female circumcision' was highlighted as an important source of support by several participants. This document is signed by official authorities in the Netherlands to support parents in resisting pressure to have FGM/C performed when traveling abroad. Also, participants' knowledge and awareness about various aspects of FGM/C and especially their negative experiences with FGM/C could be regarded as helpful to resist pressure and to protect their daughters.

## Limitations

As with many field studies, this study has limitations. First, despite our efforts to establish a rich and varied sample, our recruitment strategy may have resulted in over-representation of migrants with less favorable attitudes towards FGM/C. Specifically, the fact that some participants were recruited through a trusted intermediary with a clear position against FGM/C may have created a sample bias. However, given the difficulties inherent in research on the topic of FGM/C, the current study does provide much needed information on FGM/C among little studied migrant groups in the Netherlands.

Second, due to sensitivity of the topic and the illegal status of the practice in the Netherlands, participants may have provided socially desirable answers during the FGDs and interviews. Nonetheless, although migrants Kawous et al. BMC Women's Health (2022) 22:405 Page 10 of 11

may be unwilling to disclose their intention to perform FGM/C, data on attitudes regarding FGM/C may be more reliable than intention [24].

Third, although TPB is the most prominent theories in behavior change and it combines elements from theories of individual behavioral change and of theories that regard FGM/C as a social convention, actual behavior may be influenced by more factors than included in the TPB, so the application of this theoretical framework may have limited our views. Nonetheless, our aim was to explore the constructs of the TPB rather than to examine the predictive value its constructs have in relation to migrants' intentions to have FGM/C performed on their daughters.

# **Conclusions and recommendations**

Our findings showed that the study participants had no intention to perform FGM/C on their daughters. As a consequence of acculturation process, interest in the practice of FGM/C could wane following migration. Nonetheless, some pressure to perform FGM/C still exists after migration. It is imperative for healthcare professionals involved in the care and/or prevention of FGM/C to develop cultural competence and cultural sensitivity in the face of an increasingly diverse migrant population. Professionals should not only address the medical needs of women with FGM/C, but also broaden their scope to include their psycho-sexual and social needs. Specifically, professionals should pay more attention to family dynamics, as social pressure to perform FGM/C could be an important stressor for migrants. As their daughters grow, some parents may succumb to social pressure; therefore, they may require support in facing this pressure while attempting to safeguard their family bonds.

More research is needed to better understand how attitudes towards FGM/C wane following migration, and factors influencing the decline in the practice. Much could be gained from educational interventions with an emphasis on empowering vulnerable groups such as newly arrived migrants in facing social pressure while attempting to safeguard their family bonds. Individuals are more likely to resist social pressure and deviate from a particular social norm if they perceive more behavioral control and when their personal agency is strong [48]. The current study was not able to determine how empowered individuals differed from those less empowered individuals to protect their daughters. Future research is needed to thoroughly explore perceived behavioral control in relation to FGM/C.

### **Abbreviations**

FGD: Focus-group discussion; FGM/C: Female genital mutilation or cutting; TPB: Theory of Planned Behavior.

#### Acknowledgements

The research team extends their gratitude to Kaveh Bouteh for his research assistance and Zahra Naleie for her assistance with the data collection. We also would like to thank all the participants in this study who shared their valuable experiences with us.

#### **Author contributions**

RK contributed to the conception and design of the study, acquisition of the data, data analyses and interpretation of the data, drafting of the manuscript, and gave final approval for submission of the manuscript. MvdM contributed to the conception and design of the study, drafting of the manuscript, and gave final approval for submission of the manuscript. EA, AvdK, DG and MB carried out the data collection. All authors (EA, AM, AvdK, DG, MB and MvdM) contributed to the analysis and interpretation of the data and contributed to the drafting and revision of the article. All authors have read and approved the manuscript.

#### **Funding**

This research was funded by the Ministry of Health Welfare and Sport of the Netherlands (VWS, Grant no 326523), (https://www.government.nl/minis tries/ministryof-health-welfare-and-sport). The funder had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

# Availability of data and materials

Data, the anonymous transcripts used during the current study are available from the corresponding author upon a reasonable request that does not contravene the informed consent forms signed by the participants.

#### **Declarations**

### Ethics approval and consent to participate

The present study was conducted according to the principles of the Declaration of Helsinki. All participants received written and oral information about the study and signed a written informed consent form and gave a permission to audio-record the interviews before the interviews. This study was approved by the Medical Research Ethics Committee of the Erasmus Medical Centre in Rotterdam (Application nr. MEC-2018-2216).

### Consent for publication

Not applicable

### **Competing interests**

The authors have declared that no competing interests exist.

### Author details

<sup>1</sup>Erasmus University Medical Centre, Rotterdam, The Netherlands. <sup>2</sup>Pharos, Dutch Centre of Expertise on Health Disparities, Utrecht, The Netherlands. <sup>3</sup>Middelburg Human Rights Law Consultancy, Vinkeveen, The Netherlands. <sup>4</sup>KIT Royal Tropical Institute, Amsterdam, The Netherlands. <sup>5</sup>Marthine Bos Consultancy in Intercultural Competence, Deventer, The Netherlands. <sup>6</sup>Radboud University Medical Centre, Nijmegen, The Netherlands.

Received: 17 April 2022 Accepted: 9 September 2022 Published online: 05 October 2022

### References

- World Health Organization. Eliminating female genital mutilation: an interagency statement-OHCHR, UNAIDS, UNDP, UNECA: UNESCO, UNFPA, UNHCR, UNICEF, UNIFEM, WHO, WHO. 2008. http://whqlibdoc.who.int/ publications/2008/9789241596442\_eng.pdf?ua=1. Accessed 12 Jun 2019.
- END FGM, END FGM US, EQUALITY NOW. Female genital mutilation/cutting: a call for a golbal response. 2020. https://d3n8a8pro7vhmx.cloud front.net/equalitynow/pages/2280/attachments/original/1587032303/ FGM\_Global\_-\_ONLINE\_PDF\_VERSION\_-\_07.pdf?1587032303. Accessed 23 Feb 2020.
- Monagan SL. Patriarchy: Perpetuating the Practice of Female Genital Mutilation. PLoS ONE. 2010;2:160–81.

- UNICEF. Female Genital Cutting: A Global Concern. 2016. https://data. unicef.org/wp-content/uploads/2016/04/FGMC-2016-brochure\_250.pdf. Accessed 5 Jul 2019.
- United Nations Population Fund. Bending the Curve: FGM Trends We Aim To Change. 2018.
- Berg RC, Underland V, Odgaard-Jensen J, Fretheim A, Vist GE. Effects of female genital cutting on physical health outcomes: A systematic review and meta-analysis. BMJ Open. 2014;4:1–12.
- World Health Organization. WHO guidelines on the management of health complications from female genital mutilation. 2016. https://apps. who.int/iris/bitstream/handle/10665/206437/9789241549646\_eng.pdf. Accessed 12 Apr 2019.
- World Health Organization. Female genital mutilation: Key facts. 2018. https://www.who.int/news-room/fact-sheets/detail/female-genital-mutil ation. Accessed 1 May 2019.
- Vloeberghs E, Van Der Kwaak A, Knipscheer J, Van Den Muijsenbergh M. Coping and chronic psychosocial consequences of female genital mutilation in the Netherlands. Ethn Health. 2012;17:677–95.
- Elnashar A, Abdelhady R. The impact of female genital cutting on health of newly married women. Int J Gynecol Obstet. 2007;97:238–44.
- 11. Applebaum J, Cohen H, Matar M, Rabia YA, Kaplan Z. Symptoms of posttraumatic stress disorder after ritual female genital surgery among bedouin in israel: myth or reality? Arch Sex Behav. 2008;10:453–6.
- Williams-Breault BD. Eradicating female genital mutilation/cutting: human rights-based approaches of legislation, education, and community empowerment. Heal Hum Rights J. 2018;20:223–34.
- Njue C, Karumbi J, Esho T, Varol N, Dawson A. Preventing female genital mutilation in high income countries: A systematic review of the evidence. Reprod Heal. 2019:16.
- 14. Kawous R, van den Muijsenbergh METC, Geraci D, Van Der KA, Leye E, Middelburg A, et al. The prevalence and risk of Female Genital Mutilation / Cutting among migrant women and girls in the Netherlands: An extrapolation method. PLoS ONE. 2020;15:1–16.
- Pharos. The Dutch chain approach. 2021. https://www.pharos.nl/english/ female-genital-mutilation/the-dutch-chain-approach/. Accessed 10 May 2020
- The State Secretary of Health, Welfare and Sport the M of S and J. Statement opposing female circumcision. 2019. https://www.pharos.nl/wpcontent/uploads/2020/05/Engelse\_versie\_Verklaring\_tegen\_meisjesbes nijdenis.pdf. Accessed 1 Dec 2020.
- Amsterdam University Press. Migration from the Middle East and North Africa to Europe: Past Developments, Current Status and Future Potentials. Amsterdam, the Netherlands: Amsterdam University Press; 2014.
- UNICEF. Female genital mutilation/cutting: What might the future hold? 2014;1–6. https://www.unicef.org/media/files/FGM-C\_Report\_7\_15\_ Final\_LR.pdf. Accessed 6 Apr 2019.
- Ajzen I. The theory of planned behavior. Organ Behav Hum Decis Process. 1991;50:179–211.
- Mackie G, Lejeune J. Social dynamics of abandonment of harmful practices: a new look at the theory. 2009.
- Edwards RW, Jumper-thurman P, Plested BA, Oetting ER, Swanson L. Community readiness: research to practice. J Community Psychol. 2000;28:291–307.
- Oldenburg B, Glanz K. Diffusion of innovations. In: Glanz K, Rimmer BK, Viswanath K, editors. Health behavior and health education: theory, research and practice. London: Wiley; 2008.
- 23. Barrett HR, Brown K, Alhassan Y, Leye E. Transforming social norms to end FGM in the EU: an evaluation of the REPLACE Approach. Reprod Health. 2020;17:1–16.
- Johansen REB. The applicability of the theory of planned behavior for research and care of female genital cutting. Arch Sex Behav. 2020. https://doi.org/10.1007/s10508-020-01716-9.
- 25. Upvall MJ, Mohammed K, Dodge PD. Perspectives of Somali Bantu refugee women living with circumcision in the United States: a focus group approach. Int J Nurs Stud. 2009;46:360–8.
- 26. Gele AA, Kumar B, Hjelde KH, Sundby J. Attitudes toward female circumcision among somali immigrants in Oslo: a qualitative study. Int J Womens Health. 2012;4:7–17.
- 27. Vaughan C, White N, Keogh L, Tobin J, Ha B, Ibrahim M, et al. Listening to Victorian communities about female genital cutting. BJOG Int J Obs Gynaecol. 2015;122:104–5.

- 28. Elmir R, Schmied V, Jackson D, Wilkes L. Interviewing people about potentially sensitive topics. Nurse Res. 2010;9:12–6.
- Tong A, Sainsbury P, Craig J. Consolidated criteria for reporting qualitative research (COREQ): a 32-item checklist for interviews and focus groups. Int J Qual Heal Care. 2007;19:349–57.
- 30. Barbour R. Checklists for improving rigour in qualitative research: a case of the tail wagging the dog? BMJ. 2001;322:1115–7.
- Mohammed ES, Seedhom AE, Mahfouz EM. Female genital mutilation: current awareness, believes and future intention in rural Egypt. Reprod Health. 2018:15:1–10.
- 32. Pashaei T, Ponnet K, Moeeni M, Khazaee-Pool M, Majlessi F. Daughters at risk of female genital mutilation: examining the determinants of mothers' intentions to allow their daughters to undergo female genital mutilation. PLoS ONE. 2016;11:1–13.
- 33. Morison LA, Dirir A, Elmi S, Warsame J, Dirir S. How experiences and attitudes relating to female circumcision vary according to age on arrival in Britain: a study among young Somalis in London How Experiences and Attitudes Relating to Female Circumcision Vary According to Age on Arrival in Britain: A. Ethn Health. 2004;9:75–100.
- Johnsdotter S, Moussa K, Carlbom A, Aregai R, Essen B. "never my daughters": A qualitative study regarding attitude change toward female genital cutting among ethiopian and eritrean families in Sweden. Health Care Women Int. 2009;30:114–33.
- 35. Gele AA, Johansen EB, Sundby J. When female circumcision comes to the West: Attitudes toward the practice among Somali Immigrants in Oslo. BMC Public Health. 2012;12:1.
- Isman E, Ekéus C, Berggren V. Perceptions and experiences of female genital mutilation after immigration to Sweden: an explorative study. Sex Reprod Healthc. 2013;4:93–8.
- Vogt S, Charles E, Fehr E. The risk of female genital cutting in Europe: Comparing immigrant attitudes toward uncut girls with attitudes in a practicing country. SSM Popul Health. 2017;3 February:283–93.
- Gele AA, Sagbakken M, Kumar B. Is female circumcision evolving or dissolving in Norway? A qualitative study on attitudes toward the practice among young Somalis in the Oslo area. Int J Womens Health. 2015;7
  November:933–43.
- Exterkate M. Female Genital Mutilation in the Netherlands Prevalence, incidence and determinants. 2013. https://www.pharos.nl/wp-conte nt/uploads/2018/10/Female\_Genital\_Mutilation\_in\_the\_Netherlands\_ Rapport\_Pharos.pdf. Accessed 3 Apr 2019.
- Johnsdotter S. Created by God: How Somalis in Swedish Exile Reassess the Practice of Female Circumcision. 2002;1–291.
- Newton RJ, Glover J. "I Can't Blame Mum": A Qualitative Exploration of Relational Dynamics in Women With Female Genital Mutilation (FGM) in the United Kingdom. Violence Against Women. 2021. https://doi. org/10.1177/1077801221994913.
- Berry JW. Acculturation: Living successfully in two cultures. Int J Intercult Relations. 2005;29(6):697–712.
- Brown E, Porter C, Options PEER Unit. Evaluation of FGM Prevention among Communities Affected by FGM: A Participatory Ethnographic Evaluation Research (PEER) Study. London; 2016.
- 44. UNICEF. Female Genital Mutilation/Cutting: A statistical overview and exploration of the dynamics of change. New York; 2013.
- Johnsdotter S, Mestre i Mestre RM. Female Genital Mutilation in Europe: an analysis of court cases. 2015.
- Koukoui S, Hassan G, Guzder J. The mothering experience of women with FGM/C raising "uncut" daughters, in Ivory Coast and in Canada. Reprod Health. 2017;14:51.
- European Union. European Institute for Gender Equality. Estimation of girls at risk of female genital mutilation in the European Union Belgium, Greece, France, Italy, Cyprus and Malta: Report. 2018.
- Petit V, Tamar Naomi Zalk. Everybody wants to belong: A practical guide to tackling and leveraging social norms in behavior change programming. 2019.

# **Publisher's Note**

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.